Author’s response to reviews

Title: Study protocol for a randomised controlled trial of internet-based cognitive-behavioural therapy for obsessive-compulsive disorder

Authors:

Michael Kyrios Professor (mkyrios@swin.edu.au)
Maja Nedeljkovic Dr (mnedeljkovic@swin.edu.au)
Richard Moulding Dr (richard.moulding@deakin.edu.au)
Britt Klein Professor (b.klein@federation.edu.au)
David Austin Assoc. Professor (david.austin@deakin.edu.au)
Claire Ahern Dr (cahern@swin.edu.au)

Version: 2
Date: 9 June 2014

Author’s response to reviews: see over
3 June 2014

Editorial Board
*BMC Psychiatry*
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

**Re: Revision for Study Protocol**

To whom it may concern,

Thank you for feedback regarding required changes to the manuscript (MS: 9516330361304969) Study protocol for a randomised controlled trial of internet-based cognitive-behavioural therapy for obsessive-compulsive disorder.

As requested, the following have been reviewed:

- Competing interests reviewed and moved to the correct section
- Author’s contributions changed and moved to the correct section
- Acknowledgements changed and moved to the correct section
- Date that trial was registered recorded next to trial registration number
- The requested ethical and funding approval documentation has been forwarded to BMCSeriesEditorial@biomedcentral.com

We thank you again for your consideration of our manuscript.

Yours Sincerely,

Michael Kyrios, Ph.D., on behalf of the authors

Swinburne University of Technology
PO Box 218, Hawthorn, Victoria 3122 Australia
Tel: + 61 3 9214 4886;
Fax: + 61 3 9214 5525;

Prof Michael Kyrios: mkyrios@swin.edu.au
Dr Maja Nedeljkovic: mnedeljkovic@swin.edu.au
Dr Richard Moulding: richard.moulding@deakin.edu.au
Prof Britt Klein: b.klein@federation.edu.au
Assoc Prof David Austin: david.austin@deakin.edu.au
Dr Claire Ahern: cahern@swin.edu.au